Author's response to reviews

Title: Identification of Pneumonia and Influenza Deaths using the Death Certificate Pipeline

Authors:
Kailah Davis (kailah.davis@utah.edu)
Catherine Staes (Catherine.staes@hsc.utah.edu)
Jeff Duncan (jduncan@utah.gov)
Sean Igo (sean.igo@utah.edu)
Julio C Facelli (julio.facelli@utah.edu)

Version: 3 Date: 6 February 2012

Author's response to reviews:

Dear Editor,

Thanks for your e-mail conditionally accepting our paper (Identification of Pneumonia and Influenza Deaths using the Death Certificate Pipeline).

We have considered the latest referee’s comments and changed the manuscript as follows:

In pg. 8 we have clarified the meaning of “had a processing rate” as requested.

We believe that the Figure captions are adequate and that it is not necessary to add more details.

Pg. 16: We added this material at the request of a previous reviewer, so we will prefer to keep this material in the paper.

Table 2 has been shortened as suggested by the referee.

We have corrected all the grammatical points listed by the referee.

Sincerely,

Julio Facelli